Author's response to reviews

Title: Preferred intensity exercise for adolescents receiving treatment for depression: A pragmatic randomised controlled trial

Authors:

Tim Carter (timothy.carter@nottingham.ac.uk)
Boliang Guo (guo.boliang@nottingham.ac.uk)
David Turner (David.A.Turner@uea.ac.uk)
Iaonnis Morres (iomorres@pe.uth.gr)
Elizabeth Khalil (Elizabeth.khalil@nottingham.ac.uk)
Emily Brighton (emilybrighton@hotmail.co.uk)
Marie Armstrong (marie.armstrong@nottshc.nhs.uk)
Patrick Callaghan (patrick.callaghan@nottingham.ac.uk)

Version: 5 Date: 29 September 2015

Author's response to reviews: see over
Dear editor

Thank you for your feedback. Please find a point-by-point response to the requests below:

1. The date of trial registration has been included next to the Trial Registration Number
2. The CONSORT checklist has been completed and added as Figure 1. You also requested a consort flow diagram; this was already included (see Figure 2).
3. Information on informed consent has been included on pages 151-154

Kind regards

Tim carter